

# Growth, Technological Change and Output Gap in Russia

Panayotis G. Michaelides and George Economakis and John G. Milios and Leonidas Maroudas and Vassilis Aggelis

2005

Online at https://mpra.ub.uni-muenchen.de/74479/ MPRA Paper No. 74479, posted 3 November 2016 13:27 UTC

# Growth, Technological Change and Output Gap in Russia

by:

Panayotis G. Michaelides National Technical University of Athens

> George E. Economakis University of the Aegean

John G. Milios<sup>\*</sup> National Technical University of Athens

> Leonidas Maroudas University of the Aegean

> Vassilis Aggelis University of the Aegean

> > August 2005

<sup>&</sup>lt;sup>\*</sup> Contact author: Dr. John Milios, Associate Professor of Political Economy, Laboratory of Applied Economics, Department of Humanities, Social Sciences and Law, Faculty of Applied Mathematics and Physical Sciences, National Technical University of Athens, email: jmilios@hol.gr

# Growth, Technological Change and Output Gap in Russia

P.Michaelides, G.Economakis, J.Milios, L.Maroudas, V.Aggelis

#### Abstract

The present paper uses the "growth accounting" methodology to estimate technological change, in an attempt to formulate an explanation of Russia's economic decline and signs of recovery in the 1992-1999 time span, in relation to technological change. Also, the potential output and output gap is estimated using a Cobb Douglas production function and a Hodrick-Prescott filter for the Russian Economy before 2000. The results do show that, despite the general economic collapse during the 1990s, rooted in the Soviet economy's very structure, the level of technology has, practically, remained unchanged which, in turn, prevented the Russian Economy from further deteriorating. Meanwhile, the relationship between output gap and inflation is briefly examined and the results suggest that output gap is sufficient to explain the largest part of inflation in the Russian economy. As a final conclusion, technological progress and, consequently, long-term economic growth is linked to Russia's political and structural development.

#### J.E.L. Classification: P26, O10

Keywords: growth accounting, output gap, Cobb-Douglas, HP filter, technology, Russia

## **1. Introduction**

Russia's transition from a centrally planned to a Western type capitalist economy was marked, as known, by a rapid decline in nearly all economic indices, during most of the 1990s. Also, contrary to most former "socialist" countries in Europe, where the negative rates of change of G.D.P. and of industrial output were reversed to positive since 1994, in Russia the deterioration of the economic situation continued (Campos and Coricelli 2002, Blanchard 1997, Millios 2001, I.M.F. 2001, 2002) for a few more years until the end of the decade.

The purpose of the present paper is dual: on one hand, the estimation of change in the technological level of the Russian Economy, which constitutes a very important determinant of long-term economic growth, and, on the other hand, an attempt to better understand the supply side of the Russian economy by examining the concept of potential output and output gap.

According to Mokyr the economic importance of technology and innovation is great: "The difference between rich nations and poor nations is not [...] that the rich have more money than the poor, but that rich nations produce more goods and services. One reason they can do so is because their technology is better; that is, their ability to control and manipulate nature and people for productive ends is superior" (Mokyr 1990: preface). However, although technology can be viewed from an "econometric" point of view as an (e.g. exogenous) independent variable determining the output, it is also country specific, that is it depends on the overall complex socio-economic framework of the historically specific social formation under investigation. Our study refers to the Russian economy, as a whole, for the time period 1992-1999, which, practically, marks the end of the crisis period (Stikuts 2003) and is based on the *Growth Accounting* methodology.

On the other hand, potential output is an unobservable variable, which is related to the changes in technology. It cannot be estimated directly, but it can be estimated with the aid of several statistical and theoretical methods. Statistical methods eliminate cyclical fluctuations from the actual output time series. The statistical methods include the time trend approach and the Hodrick-Prescott (HP) filter approach. To apply statistical methods, no other additional variables than actual output are needed. This is the reason why statistical methods are widespread. On the other hand, statistical methods have several drawbacks, such as low estimation and forecasting efficiency, particularly when made over longer horizons (Stikuts 2003). The application of these methods requires relatively extensive time series.

The most essential drawback of this approach, however, is that substantial changes in the economic structure due to which the level of potential output may change – and therefore be inconsistent with the forecast –, are not taken into account (Stikuts 2003). This drawback is particularly noteworthy for the case of Russia and other transition economies, as with the advance of market economy in the early 1990s a great number of changes in the economic structure took place, and the level of Russia's potential output changed alongside with them.

Because of the several drawbacks of the statistical methods, the analysis based on the production function is used as an alternative method for measuring potential output. The most widely applied structural method is the estimation of the production function in the form of the Cobb-Douglas (CD) production function. Potential output thus estimated takes into account the changes in the economic structure. The paper is structured as follows: section 2 analyses the performance of the Russian economy during the last decade; section 3 presents the methodological framework; section 4 sets out the data and the variables used; section 5 presents and analyses the empirical results; in section 6 a further discussion of the paper's findings is set forth; finally, section 7 concludes the paper.

### 2. The Russian Economy in the 1990s: An Overview

The decentralization of the enterprise decision-making mechanisms and accountability, the price liberalization and, finally, the largest enterprise privatization in history (Boycko, Shleifer and Vishny 1995) constituted the key elements of the economic reform in the process of Russia's transition form a centrally planned economy to a Western type economy.

After the break-up of the Union, the state revenues were drastically cut. Accordingly were reduced the social benefits, which were provided directly by the state and the large state enterprises to their employees (e.g. health services, education, etc). Also, the commercial transactions among the F.S.U. countries became "foreign trade", and the COMECON collapsed. Furthermore, income taxation from non-state activities constituted a significant cost factor and drove to tax evasion and to delays in tax payments from certain large enterprises (O.C.D.E. 1997). Therefore, the (money) economy was partly replaced during the 1990s by non-monetary transactions<sup>1</sup>, which favored the spreading of tax evasion (O.C.D.E. 1997, Kaitila 2003). Meanwhile, the state was unable to control many other fundamental economic and social variables, while the quality of state education, health and transport services deteriorated significantly.

The first apparent result of the "transition" process was the dramatic output decline until 1998, combined with the very high inflation rates. As illustrated in *Table 1*, according to the official Russian statistics (Goskomstat), Russia's G.D.P decreased in the time period 1989-1994 by more than 50% of its 1989 value (which means the middle of the crisis period), and approximately by 30% during the last decade, given the recent recovery of economic activity.

However, we should, at this point, note the following fact: The official Russian statistics overestimated the prior to 1991 values of the macroeconomic data (O.E.C.D. 1995). Thus, based on the most resent studies of Gavrilenkov and Koen (1995), Kubowina (1996), Kubowina and Gavrilenkov (1997) and O.C.D.E. (1997), aggregate output decline in Russia in the 1989-1994 time span is, approximately, 35%. In these studies, an effort was made to account for various factors, which might have affected the estimations (Haaparanta and Kerkela 2000), e.g. the extent of tax evasion. Finally, under the new conditions, the small enterprises "hide" part of their income and, consequently, the aggregate output of the economy is underestimated (Milios 2001). In this framework, the extensive diffusion of non-monetary transactions should also be taken into account. The opposite is in force during the period of the recovery, i.e. after 1998. The reported increase in output is partly deceptive, because of the recession of the non-monetary transactions (Broadman 2001). This situation sets limitations to the estimations of the present paper, as far as both, the extent of the crisis of the Russian economy and the observed recovery are concerned.

<sup>&</sup>lt;sup>1</sup> For the significance of "the widespread use of money substitutes" in Russian economy see Maroudas and Rizopoulos (2002) and Aukutsiovek (1998).

Έτος	<b>'90</b>	<b>'</b> 91	·92	<b>'93</b>	<b>'</b> 94	·95	<b>'</b> 96	·97	<b>'98</b>	·99	<b>'</b> 00 <b>'</b>
Real GDP	-2.1	-12.9	-18.5	-12	-15	-4	-7.7	-1.3	3.6	6.0	5.7
(%change)											
Cons.Prices	5	93	1354	915	320	200	48	15	28	86	21
(%change)											
Investment	n.a.	n.a.	n.a.	-26.4	-12.8	-7.7	-8.2	-9.3	-6.6	-1.4	18.8
(% change)											
Exports (bn\$)	48.8	53.2	42.4	43.7	67.8	82.9	90.6	89.0	74.9	75.7	105.6
Imports (bn \$)	50.2	44.5	35	34.1	50.5	62,6	68.1	79.1	63.8	43.6	49.1
Tr.Balance	-1.4	8.7	7.4	9.6	17.4	20.3	22.5	22.5	11.1	32.1	56.5
(bn \$)											
Unemp/ment	n.a.	0.1	0.8	5.7	7.5	8.9	9.9	11.3	13.3	14	n.a.
_(%)											

Table 1: Main Economic Indicators of the Russian Federation (1990-2000)

Sources: I.M.F (1994, 2001, 2002a), OECD 1995, United Nations 1996, Express 1996

Given our reservations concerning the accuracy of the data, there was a much greater decline in industrial output (as to the total output – G.D.P.), while investments practically collapsed (Kaitila 2003, I.M.F. 2002a,b, O.E.C.D. 1995: 3 ff). Finally, the decrease in production was accompanied by a significant reduction in the Research and Development (R&D) expenditures (Goskomstat 1997).

However, the aggregate output decline did not affect all the branches of the Russian economy equally resulting in significant sectoral restructuring, which benefited the service sector at the expense of the industry's share in G.D.P. (Milios 2001). In the industrial sector, electric energy and metallurgy achieved the highest share increases, whereas the light industry and machine building, traditionally "technology intensive" sectors, were the most negatively influenced sectors (O.E.C.D. 1995: 4). Furthermore, the high inflation rates prevented the national currency from functioning as a means of storage of value and resulted in the abandonment of the ruble for international transactions, in favor of the U.S.A. dollar (I.M.F. 1994: 71).

Despite the dramatic decline of G.D.P. and of industrial production until 1998 (see Maroudas 2001: 62), the rate of the reported unemployment in Russia continued to remain low, i.e. bellow 10%, until the mid 1990s. More specifically (see *Table 1*), the rate of unemployment rose from 5.7% in 1993, to 8.9% in 1995. However, the fact that official figures underestimate real unemployment rates, or that early retirement of the elderly as well as a decline in women's participation rates in total employment help holding unemployment rates down, is not enough to explain why production collapse did not result in a rapidly increasing unemployment (Milios 2001).

As is illustrated in *Table 1*, the situation was quite different in the following years and the rate of unemployment "climbed" up to a high 14% in 1999. This behavior goes hand in hand with the positive rates of change in output (since 1998) and in investment (since 1999). The low unemployment rate did not avert the substantial decline of the living standards of the Russian population (see O.E.C.D. 1995: 125, 128-129).<sup>2</sup> The deterioration of the living standards of the Russian population is also associated with the increasing income and social inequality among the population<sup>3</sup> (Milios 2001).

 $<sup>^{2}</sup>$  For a study on the factors that affected the deterioration of the living standards of the majority of the Russian population see Milios (2001).

<sup>&</sup>lt;sup>3</sup> With an average per capita G.D.P. not exceeding the amount of 1.500 U.S. dollars yearly, the share of the other (than labour) means of income raised in just one year (1993-94) from 20,1% to 34,5%, while the wealthier 20% of the population enjoyed, in 1994, the 50% of the aggregate income (Milios 2001).

#### 3. Methodological Framework

The empirical investigation will be based, partly, on the *Growth Accounting* approach. Growth accounting was pioneered by Abramovitz (1956) and Solow (1957) and aimed at explaining the determinants of growth worldwide, after World War II. In growth accounting (see e.g. Romer 1996: 26-33), growth in a single country is decomposed over time, using a production function, into a part explained by growth in factor inputs and another part (i.e. the Solow residual), which is attributed to technological change, and is called Total Factor Productivity (T.F.P.). The basic framework can be extended in other ways (see e.g. Denison 1967, Mankiw, Romer and Weil 1992), the most common of which is to consider different types of capital and labour (Romer 1996: 26). Growth accounting has been applied to numerous cases in the last two decades (see e.g. Page 1994, Young 1994, Jorgenson 1988, Griliches 1988, Baily and Gordon 1988, Denison 1985, etc) with very satisfactory results.<sup>4</sup>

The most commonly used production function in empirical investigations using aggregate data is the Cobb-Douglas production function (Thirlwall 2001: 181). We, thus, assume a Cobb-Douglas production function with two inputs, capital and labour and Hicks-neutral technological progress.<sup>5</sup> So production at time t is given by:

$$Y(t) = A(t) \bullet L(t)^{a} \bullet K(t)^{\beta} \quad (1)$$

The notation is standard: Y is output, L labour, K capital, A the level of technology, while  $\alpha$  and  $\beta$  are the elasticities of output with respect to labour and capital, respectively.

A central problem in examining technological change and one that makes it difficult to define or characterize it is that it takes many different forms (Rosenberg 1982: 3). The most useful common denominator underlying its multitude of forms is that it constitutes any change in the application of knowledge that can make it possible to produce (i) a greater volume of output from a given amount of resources (ii) a qualitatively superior output, or (iii) a completely new output (Mokyr 1990: 6, Rosenberg 1982: 3). Technology constitutes a very crucial determinant of an economy's total productivity and competitiveness (O.E.C.D. 1996), however its direct quantification is difficult and it is often estimated indirectly using a production function.

From equation (1), using simple mathematics, we get that (see e.g. Thirlwall 1999: 181):

$$\frac{\partial Y(t)}{\partial t} \cdot \frac{1}{Y(t)} = \frac{\partial A(t)}{\partial t} \cdot \frac{1}{A(t)} + \alpha \cdot \frac{\partial L(t)}{\partial t} \cdot \frac{1}{L(t)} + \beta \cdot \frac{\partial K(t)}{\partial t} \cdot \frac{1}{K(t)}$$
(2)  
and

<sup>&</sup>lt;sup>4</sup> For instance, Young (1994) used the growth accounting methodology to argue that rapid growth of Taiwan, Singapore, South Korea and Hong Kong was mainly due to increasing labour-force and investment, and not to technological progress. Also, growth accounting has been extensively used for the study of the slowdown in productivity in the United States since the 70's.

<sup>&</sup>lt;sup>5</sup> The assumption of (Hicks-) neutral technological progress is, according to the empirical literature, a very reliable one (Thirlwall 2001: 187).

$$\frac{\partial A(t)}{\partial t} \cdot \frac{1}{A(t)} = \frac{\partial Y(t)}{\partial t} \cdot \frac{1}{Y(t)} - \alpha \cdot \frac{\partial L(t)}{\partial t} \cdot \frac{1}{L(t)} - \beta \cdot \frac{\partial K(t)}{\partial t} \cdot \frac{1}{K(t)}$$
(3)

Equation (2) implies that the rate of change in output depends on the growth in labour and capital, and on technological change, while equation (3) allows us to estimate technological change, indirectly.

Using simple mathematics, the rates of growth of labour productivity (Y/L) and capital productivity (Y/K) respectively, are given by:

$$1 = \frac{\partial Y(t)}{\partial t} \cdot \frac{1}{Y(t)} - \frac{\partial L(t)}{\partial t} \cdot \frac{1}{L(t)} \quad (4)$$
$$k = \frac{\partial Y(t)}{\partial t} \cdot \frac{1}{Y(t)} - \frac{\partial K(t)}{\partial t} \cdot \frac{1}{K(t)} \quad (5)$$

Thus, given that, typically, the sum of the values of  $\alpha$  and  $\beta$  are set equal to unity (see e.g. Dornbusch and Fischer 1993, Thirlwall 2001, Stikuts 2003, Billmeier 2004), the Cobb-Douglas production function takes the form:

$$\frac{Y(t)}{L(t)} = A(t) \cdot \left(\frac{K(t)}{L(t)}\right)^{1-a}$$
(6)

It is also an important challenge for the economy's authorities to determine, as close as possibly, the level of potential output, the level of actual output at any given time, and the direction in which they are heading. Both the level of potential output and the output gap are estimated numbers, and therefore, there is a major uncertainty in their calculation. For the case of the Russian Federation, this uncertainty may grow larger now because the economy seems to have undergone some significant changes during the 1990s. As a result the Russian authorities should place increased weight on a range of relevant indicators to assess the degree of pressure on the economy's production capacity.

General research on output gap started with Okun (1962) and has been abundant ever since (for instance, see Kuttner 1994). A variety of methods can be used to estimate potential output and the output gap. However, roughly speaking, there exist two broadly used methods for the estimation of potential output: The HPfilter and the production function. For a review see Bolt and Els (2000). For a brief presentation of some less popular techniques see Slevin (2001). The linear, two-sided HP-filter approach is a simple and widely used method by which the long-term trend of a (macroeconomic) series is obtained using only observable, i.e. actual, data. The trend is obtained by minimizing the fluctuations of the actual data around it, i.e. by minimizing the following function:

 $\sum \left[ \ln y(t) - \ln y^{*}(t) \right]^{2} - \lambda \sum \left[ \left[ \ln y^{*}(t-1) - \ln y^{*}(t) \right] - \left[ \ln y^{*}(t) - \ln y^{*}(t-1) \right] \right]^{2}$ 

where  $y^*$  is the long-term trend of the variable y and the coefficient  $\lambda$  determines the smoothness of the long-term (output) trend, expressing the potential output in this case.

The HP-filter approach has two positive features (Stikuts 2002): First, the obtained trend is influenced by shocks. Second, it is simple to measure. However, the HP-filter alone produces a good result only when data on a relatively stable economic environment are used. In this case, the HP-filter along with the relevant and econometric methods has to be used instead of simple linear trend.

Numerous shocks affect economic growth in developing countries, and substantial and accelerated changes in actual output do not necessarily signal either expansion or contraction of potential output. In this context, reliance on the HP-filter approach *alone* may lead to erroneous conclusions. In addition, sources often subject to criticism such as biases at the ends of the time series may influence the economic policymakers' decisions. Irrespective of the given drawbacks, the HP-filter approach is widely employed because of its simplicity.

This method, in contrast to the production function approach, does not use information provided by the factors of production, such as capital stock, workforce and technological development. It does not measure the influence of structural shocks on potential output and hence the output gap estimation may sometimes be biased. In order to avoid it, the estimation of the production function is used as an alternative output gap estimation approach.<sup>6</sup> Practically, its most important advantage lies in the possibility to account explicitly for different sources of growth (Billmeier 2004).

This method estimates a production function where real G.D.P. is some function of capital, labour and technology. The production function is then estimated when the capital stock is being fully utilized and the labour force is fully employed. This method has been used by various researchers (see Artus 1977, Giorno et *al.* 1995, De Masi 1997, Bolt and Els 2000, Senhadji 2000, Slevin 2001, etc.) Continuously, HP filter smoothing techniques have been used in the production function approach to filter technical progress and potential employment (see Giorno et *al.* 1995, Bolt and Els 2000, Fagan et *al.* 1998).

The production function may take various forms, yet the most widely used is the Cobb-Douglas (CD) production function specification.<sup>7</sup> As seen, the CD function may be written as follows:

$$\mathbf{Y}_{t} = \mathbf{A}_{t} \mathbf{L}_{t}^{\alpha} \mathbf{K}_{t}^{1-\alpha} (1)$$

where  $Y_t$  denotes output at constant prices,  $K_t$  denotes capital stock at constant prices,  $L_t$  denotes the number of the employed persons,  $A_t$  characterizes the Total Factor Productivity (T.F.P.) and  $\alpha$  is the elasticity of production factors.

Alternatively, equation (1) can be written as:

$$\frac{\mathbf{Y}_t}{\mathbf{L}_t} = \mathbf{A}_t \left(\frac{\mathbf{K}_t}{\mathbf{L}_t}\right)^{1-a} (6)$$

<sup>&</sup>lt;sup>6</sup> For overviews of the HP filtering method shortcomings see Harvey and Jaeger (1993), King and Rebelo (1993), Cogley and Nason (1995) and Billmeier (2004).

<sup>&</sup>lt;sup>7</sup> The CD function has drawbacks as well (Stikuts 2003): First, it is a simplified reflection of reality. For instance, it considers as homogenous the production and labour expanded originating from different sectors and skills. In other words, labour force or capital of the manufacturing industry is regarded as being the same as in the banking sector and can be easily transferred from one sector to another. Second, the data employed may result in a biased estimation, as in any econometric estimation, since the application of more accurate data is restricted by irregular availability (e.g. data concerning the utilisation of capital are not accessible with adequately high frequency to be used in econometric studies). Third, natural or optimal factor utilisation capacity is difficult to define. Finally, the Solow residual is a substantial component of the production function, which is calculated as estimation residual and as such is economically unexplained and freely interpretable. Irrespective of its drawbacks, the CD function is one of the methods, which along HP-filter is widely used to estimate the potential output (Stikuts 2003). For a brief review of the model's theoretical limitations see Thirlwall (2001:185-7), which are, however, of limited practical character, as the author himself implies (see ibid :187).

Taking logs equation (6) yields a linearised form:

$$\ln\left(\frac{\mathbf{Y}_{t}}{\mathbf{L}_{t}}\right) = \ln\mathbf{A}_{t} + (1-\alpha)\ln\left(\frac{\mathbf{K}_{t}}{\mathbf{L}_{t}}\right) \quad (7)$$

This linearization reduces the number of coefficients to be estimated and eliminates the multicolinearity problem of the explanatory variables (Stikuts 2003).

Alternatively, taking *logs* equation (1) can be written as:

 $\ln Y_{t} = \ln A_{t} + \alpha \ln L_{t} + (1 - \alpha) \ln K_{t}$ (8)

Potential output  $Y_{t}^{*}$  is derived by inserting the potential values of the production factors. In other words:

 $\ln Y_{t}^{*} = \ln A_{t}^{*} + \alpha \ln L_{t}^{*} + (1 - \alpha) \ln K_{t}^{*}$  (9)

where \* denotes the potential value of the production factor.

The next step is to measure the potential value of production factors. The actual value of capital stock is used as a substitution for its potential value, as capital stock cannot fluctuate substantially, and it is assumed that the capital stock available is always used at its potential. Thus, we have that:

$$\mathbf{K}^* = \mathbf{K}_t \quad (10)$$

The Total Factor Productivity  $A_{t}^{*}$  is partly estimated by the production function as the residual  $\varepsilon_{t}$  of equation (7) and the potential level is determined by the HP-filter to obtain a smooth time series. Consequently:

$$A_{t}^{*} = A_{s_{t}}$$
 (11)

where  $A_{s_t}$  is the HP-filtered residuals of equation (3) characterising T.F.P.

Potential labour input is estimated using the NAWRU (non-accelerating wageinflation rate of unemployment) concept. The NAWRU is the unemployment rate at which wage inflation is constant. Several studies (see Layard et al. 1991) show that the equilibrium unemployment rate changes over time, but it generally follows the actual unemployment rate (due to hysteresis and labour market inelasticity).

Elmeskov's (1993) popular method is used in this paper to construct a time varying NAWRU. This approach has also been used by Bolt and van Els (1998) and Bolt (2000) to estimate the output gap in European Union (E.U.) countries and in Slevin (2001) for the case of Ireland. It is based on an equation, which relates changes in unemployment with those in wage inflation:

$$u_t - NAWRU_t = \lambda \Delta^2 w_t, \lambda < 0$$
 (12)

where  $u_t$  is the actual unemployment rate, NAWRU<sub>t</sub> is the (natural) unemployment rate, which has no effect on wage inflation and w(t) is the average gross wage in the national economy.  $\Delta$  is the first difference,  $\Delta^2$  is the second difference and  $\Delta^3$  is the third difference operator.

Taking left and right first differences of equation (12) leads to an equation for  $\lambda$ :

$$\lambda = \frac{\Delta u_t}{\Delta^3 w_t}, \Delta^3 w_t \neq 0 \qquad (13)$$

when inserting the latter (13) into equation (12) we get:

NAWRU<sub>t</sub>= 
$$u_t - \frac{\Delta u_t}{\Delta^3 w_t} \Delta^2 w_t$$
 (14)

Equation (14) implies that the NAWRU is equal to the actual unemployment rate, which is adjusted by unemployment rate changes and wage inflation relationship. The resulting series in then smoothed to eliminate erratic movements using the HP filter. Consequently, potential employment is calculated as follows:

$$L_{t}^{*} = L_{s_{t}} [1-NAWRU_{s_{t}}] (15)$$

where  $L_s$  is the HP-filtered labour time series and NAWRU<sub>s t</sub> is the HP- filtered NAWRU time series.

Substitution of the potential values of the production factors obtained from equations (10), (11), and (15) into equation (6) yields the time series of the potential output.

#### 4. Data and Variables

The significance of the factors, entering the production function of the Russian economy, is tested using the available data collected from the publications of the International Monetary Fund (I.M.F.) entitled *International Financial Statistics* (2001, and 2002a, 2004). The data available is on an annual basis and covers the period 1992-1999 (eight observations).

The econometric method used for the estimation of Russia's technological level employs annual G.D.P. data starting with the year 1992. The selection of data has been determined by several factors: First, we start with 1992 because it is the first year, after the major economic reform process in the early 1990s, that data are available. On the other hand, we stop in 1999, because, it is the year that a significant reform process causes Russia's national economy to experience structural shocks, (such as increasing balance trade, decreasing pattern of inflation, changes in terms of trade, slowly growing competitive markets, etc.), the intensity and scope of which differentiates the model estimation for the given period when compared with the prior 2000 situation (IMF 2002a, Milios 2001) and is consistent with the thesis of Stikuts (2003: p. 7) who, explicitly, stated that the Russian crisis ended in 1999. In addition, no sufficient and reliable data are available for the study of the extremely short remaining (recovery) time span and so we are prevented from attaining a statistically very satisfactory level of estimation.

Data on the capital stock at constant prices is not published. However, we estimated it using the *Perpetual Inventory* method. Data on labour or the number of employed persons is provided by the International Monetary Fund annually. As can be seen from a simple inspection of our data concerning labour, since 1992 we observe a slight decline in labour due to the shrinking number of economically active population because of demographic reasons and migration. The demand for workforce in the last decade, on the other hand, has been influenced by a number of factors. The most significant decrease in labour demand could be related to the Russian crisis. Many enterprises had to narrow production and this led to contracting labour demand. This time, however, the unemployment rate did not rise in proportion to the fall in the output growth rate, which may be seen as an indication of producers' unwillingness to dismiss "redundant" workforce completely as they were convinced that the incurred economic hardships were temporary, and, with production soon recovering, hiring of labour would be related to extra costs.

In accordance with the decreasing tendencies of capital stock, labour has decreased since 1992. This implies that resources leading to a sustainable and robust

growth in Russia currently are unavailable. Russia's economic development relies basically on an upswing in capital productivity driven by technological advances and a more intensive utilisation of capital stock rather than employing additional workforce.

#### Dependent Variable

1. The Gross Domestic Product (G.D.P.) of Russia at constant 1991 prices measured in billions of Rubles.

#### Independent Variables

- 2. Labour in Russia is measured as the number of employees in billions.
- 3. Fixed Capital Formation in Russia is estimated using the *Perpetual Inventory* method (see Appendix) and is expressed in billions of Rubles at constant 1991 prices.

### **5. Empirical Results**

As seen, the most widely used functional form of the production function is the linearised Cobb Douglas specification (Thirlwall 2001, Stikuts 2003), which reduces the number of coefficients to be estimated and eliminates the multicolinearity problem of the explanatory variables (Stikuts 2003). The relationship is, thus, linearised and we use a time-series data set for the period 1992-1999, when data is available. The results of the regression through Ordinary Least Squares (O.L.S.), which is used for the estimation of the linearised Cobb-Douglas production function (Andrikopoulos 2000: 358), are presented in *Table 2* :

$$\ln\left(\frac{\mathbf{Y}_{t}}{\mathbf{L}_{t}}\right) = \ln\mathbf{A}_{t} + (1-\alpha)\ln\left(\frac{\mathbf{K}_{t}}{\mathbf{L}_{t}}\right) \quad (7)$$

Table 2 presents the regression results for the dependent variables.

Production Function for Russia, 1992-1999							
Independent Variables	Estimate	t-statistic					
Intercept	2,35						
β	0,41	2.08*					
implied a	0,59						
$R^2$	0.42						
F-Ratio	4.33						
S.E.E.	0.21						
M.A.E.	0.16						
D.W statistic	0,70						

*Table 2*: Regression Results for the Cobb-Douglas Production Function for Russia, 1992-1999

Note: \* Significance at the 10% level

The signs of the estimated coefficients are consistent with the stated hypotheses and economic theory; the results are statistically significant for the independent variable, while the equation explains a considerable part of the variability of G.D.P. The results should be assessed as satisfactory given the various imperfections in this sort of country data (Mankiw, Romer and Weil 1992: 408), as well as given the crisis period and the various violent shocks that the Russian economy faced in the under investigation period (Stikuts 2003).<sup>8</sup>

Also, there are no signs of serious violation of the basic assumptions concerning the residuals, as was easily confirmed with the aid of the relevant procedures (see Samouel et al. 1996: ch. 12): specifically, the normality of the errors was assessed through the examination of the frequency distribution of the residuals as well as by reference to the Q-Q or P-P normality plot, which is a special type of plot for checking normality. As far as the assumption of homoscedasticity is concerned, compliance with this assumption was evaluated by examination of the scatter plot of the standardised residuals against the predicted values. Finally, as for the assumption that the residuals are independent of each other, investigation of the scatter plot of the standardized residuals against the time variable provided some idea of possible dependence between successive values, i.e. autocorrelation effect.<sup>9</sup>

Labour elasticity derived is 0.59, and the value of capital stock elasticity is 0.41. These values are, in general terms, consistent with estimations produced by researches on other countries. For instance, as known, the majority of research papers indicate that the value of labour elasticity for the developed countries is around 2/3, while that of capital is 1/3 (labour elasticity estimates in the US are within the range of 0.59 and 0.87, and from 0.57 to 0.59 in Germany) (see Bolt and Els 2000, Dimitz 2001). Recent studies show that in Estonia labour elasticity is around 0.67 (Stikuts 2003). On these grounds, the estimation of labour and capital elasticity of Russia's production function may be regarded as credible.

Potential labour (L\*) and actual labour (L) are illustrated in Figure 1.

Figure 1: Potential (L\*) and Actual (L) Labour, 1992-1999 (bn of employees)



Figure 2 illustrates the NAWRU and the actual unemployment rate (U).

<sup>&</sup>lt;sup>8</sup> The estimation of the unrestricted non-linearised Cobb-Douglas model did not yield acceptable statistical or theoretical results.

<sup>&</sup>lt;sup>9</sup> An alternative diagnostic is provided by the Durbin–Watson statistic which indicates the degree of autocorrelation in our dataset. However, given the value of this statistic in our dataset, the hypothesis that the residuals are autocorrelated cannot be accepted.



Figure 2: NAWRU and Actual Rate of Unemployment, 1992-1999 (%)

The actual level of capital stock, which is by assumption equal to its potential level, is illustrated in figure 3.



Figure 3: Potential (K\*) and Actual (K=K\*) Capital, 1992-1999 (bn Rubles)

Finally, Figure 4 illustrates the potential (*TFP*\*) and actual (*TFP*) level of TFP, expressed by the residuals of the CD production function.



Figure 4: Potential (TFP\*) and Actual (TFP) Level of TFP, 1992-1999

Substituting potential estimates for actual data and the technological variable for the estimated residual variable, Russia's potential (Y\*) and actual (Y) G.D.P. are calculated (Figure 5). Potential G.D.P. (Y\*) has decreased by about 10% per annum on average. In this case, changes in actual G.D.P. are much closer to the level of potential G.D.P., and movements in the latter are explained by changes in the production factor capacity utilisation. Actual G.D.P. has decreased by about 8% and fell behind potential G.D.P. only in the years 1995, 1996 and 1997.



Figure 5: Potential (Y\*) and Actual (Y) Output, 1992-1999 (bn Rubles)

Finally, figure 6 summarises the output gap performance.



We notice that the elasticity of production, as far as labour is concerned, is higher than the elasticity of capital, because the estimated  $\alpha = 0.59$  is greater than  $\beta = 0.41$ . Thus, production is much more "sensitive" with respect to labour than with respect to capital. These findings suggest that the Russian economy tends to be a labourintensive economy and, at a first level, it seems that this way we are offered, *ceteris paribus*, a possible explanatory parameter of the limited unemployment under the conditions of economic crisis.

In *Table 3* that follows, we can see the estimated average annual rates of change in production and inputs, labour productivity, capital productivity and total productivity (i.e. technological change), as a result of the application of equations (4), (5) and (3) respectively, after the empirical estimation of the production function.<sup>10</sup>

Table 3: Average annual rates of change	ge in production,	, labour, cap	oital, productivity of
labour, productivity of capital and total	productivity (T	.F.P.) (1992	2-1999).

· .	~	1		/	
[dY/dt]/Y	[dL/dt]/L	[dK/dt]/K	[dY/dt]/Y–	[dY/dt]/Y–	[dA/dt]/A
			[dL/dt]/L	[dK/dt]/K	
- 7.9 %	-2.3 %	- 16.9 %	- 5.6 %	9.0 %	0.4%
Source: IM	E (2001 20	()) authora?	alaboration		

Source: I.M.F. (2001, 2002a), authors' elaboration

In the time period 1992-1999, the growth rate in production was negative, and declined by -7.9% per year. The capital stock contributed to this fall with a rate equal to -16.9% per year (which was caused by the collapse of investments), while labour declined by -2.3% per year and managed to withhold the significant decline of production, given its higher share in production. Finally, the annual average rate of change in T.F.P. during the time period 1992-1999 was slightly positive and equal to

<sup>&</sup>lt;sup>10</sup> The change in T.F.P. is expressed through  $\frac{\partial A(t)}{\partial t} \cdot \frac{1}{A(t)}$  which is, as seen, attributed mainly to

technological change.

0.4%. We can see, therefore, that labour and technology constituted the "sheetanchor" of the Russian economy during the under investigation time period, since they kept the negative average annual rate of change in the G.D.P. to "only" -7.9%, when a dramatic decline of the capital stock took place in the economy.

The average annual rate of change in the productivity of capital (Y/K), during the specific time-span is high (9%) and is due to the capital's over-employment, which is the result of the rapid capital decline that takes place at a faster rate than that of product, resulting, in turns, in the production of more output by a smaller part of the capital used, as time goes by. The average annual rate of change of labour productivity (Y/L) is negative (-5.6%) and it is practically due to the limited decline of labour (-2.3%) in the under investigation time period, while the product has more than dwindled (-7.9%). This practically means that less product is being produced by a slightly less amount of labour as time goes by, and is consistent with the findings of Kaitila (2003: 15).

The above findings could be better highlighted if we take into consideration the fact that during the period under investigation (1992-1999) emerge both a dramatic crisis and a tendency towards recovery (after 1997) of the Russian economy. *Table 4* that follows isolates the crisis period (1992-1997) from the recovery period (1998-1999).

*Table 4*: Average annual rates of change in production, labour, capital, productivity of labour, productivity of capital and total factor productivity (T.F.P.) (1992-1999).

						<i>.</i>
Έτος	[dY/dt]/Y	[dL/dt]/L	[dK/dt]/K	[dY/dt]/Y-	[dY/dt]/Y-	[dA/dt]/A
				[dL/dt]/L	[dK/dt]/K	
1992-'97	-14.6%	-3.3 %	-16.2 %	-11.2 %	2.4 %	-6.0%
1992-'99	- 7.9 %	-2.3 %	- 16.9 %	- 5.6 %	9.0 %	+0.4%
	/					

Πηγή: I.M.F. (2001, 2002a), authors' elaboration

If we take a closer look at the results concerning the sub-period 1992-1997, in relation to the under investigation period 1992-1999, we believe that the following two fundamental and interrelated conclusions are drawn effortlessly. First, except for the growth rate of the capital stock - which remains almost unchanged during the period under investigation - all the other variables experience an almost dramatic deterioration, in terms of growth rates, during the crisis sub-period, since the negative change in production progresses with an almost double average growth rate (-14.6%) compared with the -7.9% of the whole period. The negative average annual rate of change in employment for the sub-period, compared with the slightly changed one for the whole period indicates that the positive change of employment progresses in parallel with the observed significant rise in the unemployment level, during the recovery period (see Table 1). Second, the sub-period of the emerging recovery (1998-1999), which is related positively to the changes in employment, in labour productivity and especially in capital productivity and, partly, to the change in the technological level, comes to blunt without however to obliterate (yet), the results of the crisis period of the Russian economy (i.e. negative growth rate in production, in capital stock, in labour productivity and, almost non existing technological progress between 1992-1999).

A symptom resulting from the latest findings is the decrease in the ratio of the (fixed) capital's value to the labour force's value for the period 1992-1999 (as can be easily computed with the aid of the data available) or, in other words the decrease in

the fixed capital's value per labour force's value unit. This indicates the relative decrease of the fixed capital's value with respect to the aggregate capital and corresponds conversely to the "economy in the use of constant capital". This finding demonstrates the reduced efficiency – as to the saving of fixed capital – of the intense-discipline methods of labour, etc. for the period under investigation.

We will insist on the subject of the technological level, which remains practically unchanged, in the period under investigation, and is responsible for about 5% of the change in G.D.P., as is evident from the data in *Table 3*.<sup>11</sup> Thus, we notice that the production decline is limited, in the long run, by the non-negative growth rate of the technological level of the Russian economy.<sup>12</sup> However, the problematic – concerning the level of technology - situation in the crisis sub-period, namely between 1992-1997, was caused by the cut down of the Research and Development (R&D) expenditures. The R&D statistics for the sub-period of crisis in Russia are overwhelming.<sup>13</sup> Additionally, from surveys in various productive branches it became evident that the Russian factories' machinery and equipment were technologically "old".<sup>14</sup> Furthermore, the poor state of the production infrastructure had a negative effect on exports (O.C.D.E. 1997) as well.<sup>15</sup> Thus, the findings of our investigation confirm the so-called "Russian paradox" (Milios 2001), expressing the fact that a reform which has advanced in the name of economic development and modernization led the country to economic and technological retrogression, or, in the words of Kagarlitsky (1995: 88): "What is unusual about the capitalist reforms in Russia is that for the first time in history, the 'old' structures are on the technological level [...] far higher than the 'new'" (see also Maroudas and Rizopoulos 2002: 126).

Finally, as far as the estimation of the potential output is concerned, this is associated with the supply side of economy and the positive output gap should be expressing excess demand and the negative output gap should be expressing excess supply. Positive output gap is associated with the inflationary pressures in the

<sup>&</sup>lt;sup>11</sup> Doyle *et al* (2001) estimated that in the last decade total factor productivity contributed 9% of G.D.P. in Slovakia, 44% in Poland, 51% in the Czech Republic, 82% in Slovakia and 122% of G.D.P. in Hungary.

<sup>&</sup>lt;sup>12</sup> It is interesting to note that in a seminal article, Wladimir Andreff (1978) followed, in general terms, a similar methodological framework, concerning the relation between the level of technology and the economic slowdown of the Eastern European countries in the 1950s and in the beginning of the 1960s. Also, for a brief survey of some classic articles measuring the percentage of growth which rises from an increase in total factor productivity in the former U.S.S.R, see Andreff (1978: 50).

<sup>&</sup>lt;sup>13</sup> In 1991 the expenditures for R&D amounted to approximately 1.85% of G.D.P. and in 1997 they were cut down to 0.5%. Namely, between 1991-1997 a rapid decline in the expenditures for R&D is observed, equal to 72.97%, which implies a decline of 19.59% annually. Until 1993, the capital funds allocated in the various scientific programs in Russia were cut down approximately four (4) to ten (10) times, in comparison with 1990. Meanwhile, more than five hundred (500) high-leveled scientists had emigrated and over seventeen thousand (17,000) had fled "for long term employment abroad". Furthermore, many of the 4,500 research centers disappeared and 1,100 became private (Milios 2001).

<sup>&</sup>lt;sup>14</sup> For example, according to O.E.C.D. (1997-8: 37-38) about half of the existing oil pipes were older than twenty (20) years of age and, as a result, approximately 2% of oil production was lost because of leakages and accidents. However the "aging" of machinery and equipment has a historical background in the past regime (see on this, among other works, Pavlovitch 1996, Goskomstat 1997, O.C.D.E. 1997, Fakiolas and Fakiolas 1999).

<sup>&</sup>lt;sup>15</sup> For example, the inoperative drillings of oil – which constitutes one of the basic export products of Russia (Analytis 1999: 299, Kaitila 2003: 8, 19) – increased from 4,000 to 32,000 in 1993, due to the shortage in appropriate capital and technological infrastructure, whereas the absence of the appropriate chemical substances for the improvement of the oil-bearing process's efficiency enhanced the problem. In general, the inefficient technology resulted in pumping half of the quantities of oil possible (Analytis 1999: 298).

economy and therefore policy makers are very concerned about it. Let us correlate the output gap with the consumer prices index for the case of Russia.

Tuble 5. Conclution of Output Oup with inflation							
	t	t-1	t-2	t-3			
Consumer Price Index	0,14	0,02	0,12	0,76			
Wage Inflation	0,14	0,05	0,02	0,55			

*Table 5:* Correlation of Output Gap with Inflation

As we can see from the correlation Table 5 the highest (linear) correlation is with lag 3, meaning that output gap transforms into inflation after about three (3) years. The contemporaneous correlation coefficient over this period of 0.76 indicates strong correlation. We note a roughly similar performance when using the two different measures of inflation, a thing which is due to the – generally – similar pattern that the two inflation measures follow. Consequently, the results indicate that there is a strong relationship between the consumer price index and the output gap. We also note that the Cobb Douglas production function output gap, which uses HP filtered T.F.P. as a measure of technology, is significant in explaining inflation in the presence of lagged inflation.

However, the length of the data series is a serious limitation towards the explanation of the empirical results. The resulting inflationary pressures are also caused by external factors, such as depreciation of the currency, high oil prices, since Russia is, now, an open economy and excess domestic demand partially results into imports growth and probably because the limited competition among importers partially excess demand influence also prices.

#### 6. Further Discussion

The main approaches to the case of the economic crisis in the countries of Eastern Europe and the former Soviet Union make use of arguments, which are based on the view that the crisis had mainly "external" causes as far as the transition process is concerned, while the factor of technology is systematically neglected (see Milios 2001).<sup>16</sup> Nevertheless, the statistical data available (see *Table 1*) and our findings that output gap fuels inflation, or in other words the "domestically generated" character of the inflationary mechanism, which locates its origin inside the Russian social formation, and only partly in external factors, seem to refute these analyses. On the contrary, it seems that the restructuring of external trade and the adoption of the international market prices improved the situation for the Russian trade, and in combination with the rapid decline in internal demand, it accelerated Russian exports (traditionally consisting of raw materials and oil-products<sup>17</sup>) to the countries of O.E.C.D. In the past, Russia exported this kind of products mainly to the COMECON countries, but the prices had been significantly lower than those of the western capitalistic markets.<sup>18</sup> The dissolution of the COMECON and the adoption of the

<sup>&</sup>lt;sup>16</sup> According to these approaches, output decline is due to the deterioration of the position of the former "socialist" countries in the global market, i.e. to "external" factors. For a political economy approach in modelling the transition process, see Marangos (2003a).

<sup>&</sup>lt;sup>17</sup> For a comparative analysis of the Russian trade fleet's performance before and after the break-up of the Soviet Union, see Economakis *et al* (2004).

<sup>&</sup>lt;sup>18</sup> For alternative transition strategies to market relations in Eastern Europe, see Marangos (2003b).

prices of the global capitalistic market improved the Russian trade balance and the balance of payments on current account in real terms, even during the period of the appreciation of the ruble since the 1993 (O.E.C.D. 1995)<sup>19</sup>.

In a first attempt to interpret the Russian social formation's economic crisis, it can be mentioned that the collapse of the Soviet regime followed by the cessation of state regulation and price liberalization for the sake of competition, allowed the "free" enterprises to adopt to high prices (a fact that led to very high inflation rates, especially during the first years of the transition process, see *Table 1*), in an effort to increase their profit margins and to cope with demand decline, caused by the drastic cut in "planned" state orders (Gorbachev 1987: 145, O.C.D.E. 1997).

However, this price explosion caused a further dramatic fall in demand. A vicious cycle of rising prices and decreasing demand sunk, thus, the Russian economy into continuous output decreases and economic crisis. Meanwhile, in the monopolistic economic structure that emerged (Maroudas 2001), the few new technologies created led to high prices, driving to further price increases. In this way, many enterprises were prevented from further buying and using new technologies, because of a lack of strong incentives due to their high costs.

In the present paper, we defend the thesis that the particularities of the Russian private business economy, which are expressed – at least until 1998 – through the coalition of the new ruling class with the employees, the preservation of the main volume of labour, the limited development of competition, the non-pursuit of profitability (as the main objective of production/reproduction) and the absence of entrepreneurial incentives for business plans, and consequently for technological innovation, formed a special historical framework for the expression of the crisis, the overcoming of which constitutes a precondition, as well as the result, of the liberation of the counterbalancing tendencies of the crisis (concerning these counterbalancing tendencies see Marx 1991: 362-364, see also Liodakis 2001: 57).

More precisely, as the stagnation of the Soviet economy perpetuated (Clarke 1993, Rutland 1985, Nove 1977), the soviet (political) leadership moved towards the encouragement of the tendency for enterprise autonomy, with a view not of their being autonomous from the "economic plan", but with the hope that competition would prevail among the "free" enterprises (Filtzer 1994). However, the privatization process was nothing more than a transfer of enterprise ownership from the central planning authorities to managers and senior cadres (Radyguine 1997, Milios 2001, O.E.C.D. 1995), who were supported by the delegates of the employees (O.E.C.D 1995, Earle and Estrin 1997, Blasi, Kroumanova and Kruse 1997, Radyguine 1997, Milios 2001, Milios 2001, Maroudas and Rizopoulos 2002).<sup>20</sup>

Specifically, the particularities of the Russian private business economy could be expressed as "a particular form of socio-economic organization" that seems to persist after the demise of the Soviet Union (see Maroudas 2001: 52, 60). This particular form of socio-economic organization consists of *four structural elements, in their unity:* First, "the creation of a paternalistic-type coalition between managerial staff and workers" (Maroudas 2001: 52). This coalition between managerial staff and workers was expressed, as seen, from the workers' point of view, through their support to the transfer of the enterprises to the existing managers and their senior

<sup>&</sup>lt;sup>19</sup> However, the external value of the ruble fell considerably in real terms during the autumn of 1998 (I.M.F. 2002a), while import prices increased correspondingly in ruble terms (O.E.C.D. 2003).

<sup>&</sup>lt;sup>20</sup> For an excellent attempt to model the privatization process in Transition Economies, see Marangos 2004.

cadres,<sup>21</sup> and from the new owners' point of view by the promise of the maintenance of the current employment levels. This ability to promise is based, in turns, on a second structural element: the limited development of competition. Both these structural elements are reduced to the particular type of privatization of the ex-Soviet enterprises (see Milios 2001). At the same time, the absence of competition development, but also the increasing negotiating power of enterprises *vis a vis* the national, regional and local power centers (see Maroudas 2001: 61-64) is connected directly with an additional, third structural element of the particular form of socioeconomic organization in Russia: The reproduction of the given status of economic authority and power, and not the creation of profit is probably the central question of production. It is a fact that, in turns, determines as a fourth main characteristic a low level of entrepreneurial incentive and – consequently of risk – for long-term investment plans (see Maroudas and Rizopoulos 2002: 127) and, therefore, for technological innovation as well.<sup>22</sup>

At this point, it is interesting to note that, to the extent that the paternalistictype coalition between managerial staff and workers sets restrictions and/or limits to the control of the first over the production process it sets, at the same time, disincentives for the introduction of new technologies, as long as technological change is also a process used by the employers in order to introduce new ways of control over labour (Bowles and Edwards 1993: ch. 11).

In this way, "[t]he monopolistic capitalism that emerged from the state's withdrawal is deprived not only of the 'planned' state markets but also of all traditional incentives,<sup>23</sup> without being able to create new ones" (Milios 2001: 82).

To the crisis factors for the period until 1998, we should add "the weakening of the state to the extent that it could clearly no longer perform its most elementary functions" (Maroudas and Rizopoulos 2002: 126), primarily of a consistent and reliable economic policy<sup>24</sup> (also see O.E.C.D. 1995: 1). The institutional deregulation is an additional factor immanently connected with the technological fall back.<sup>25</sup>

However, after 1997 the Russian economy shows signs of recovery from the dramatic crisis that it has been experiencing: In brief, the inflation rate decreased considerably, while the declining trend of the real G.D.P. was reversed to increasing, and investments, in constant prices, have increased in the specific time span (see *Table 1*). At the same time, as we have already noted, the period of recovery (1998-1999) that comes to blunt without, however, being able to obliterate the results of the crisis period of the Russian economy, is related to changes in employment (and unemployment), in labour productivity and especially in capital productivity, and of course to the change in the level of technology. It is not possible, at present, to formulate safe predictions for the overcoming of the crisis. It is also early for final

<sup>&</sup>lt;sup>21</sup> In the same line of argument, "wage arrears representing in practice interest-free loans granted from employees to their enterprise" (Maroudas and Rizopoulos 2002: 130).

<sup>&</sup>lt;sup>22</sup> "The major way for capitalists to compete within sectors is by introducing technological innovations. This move is spurred not only by the need to save on rising costs, but also by the need to improve efficiency, that is, the units of output per capital invested, and thus competitiveness and profitability" (Carchedi 2001: 79).
<sup>23</sup> "Given the monopolistic structure of the economy, the old system was bound to an inherent type of

<sup>&</sup>lt;sup>23</sup> "Given the monopolistic structure of the economy, the old system was bound to an inherent type of rationality, by redistributing income, creating incentives, setting 'production norms' and 'plan objectives', and pursuing a moderate growth" (Milios 2001: 82).

<sup>&</sup>lt;sup>24</sup> For the importance of state intervention in ensuring macroeconomic stability in transitional economies see Siriopoulos and Asteriou (2001).

<sup>&</sup>lt;sup>25</sup> Following Loasby (2002:41): "[I]nstitutions provide both the necessary baseline and the boundaries across which one may move to an adjacent state of knowledge".

conclusions concerning the causes of the observed signs of recovery of the Russian economy. Yet, we will attempt to formulate very briefly some first thoughts defending the thesis that the observed signs of recovery and their prospects should probably be looked up in the following factors and directions.

First, in the means of overcoming the crisis, which is connected to tendencies that the crisis itself liberates during its evolution; that is the intensity of competition, the massive capital destruction, the rise of the unemployment rate (sometimes despite the rise in employment volume) and the decrease in real wages.<sup>26</sup> As far as the latter three are concerned, our findings are clear and already analyzed earlier. Specifically, since the recovery is based on the ground of a continuous destruction of fixed capital – which goes along with the increase in investments after 1999 – a possible trend of "modernization" of the productive basis is implied. As for the former, that is the strengthening of competition, which constitutes a critical factor driving to a dynamic development from a capitalist point of view, we only have some first indications of the strengthening of competition<sup>27</sup> in the economy (see E.B.R.D. 1999, O.E.C.D. 1995: 27 ff, B.C.C.R. 2003 Broadman 2001, Kaitila 2003: 12).

On the basis of the arguments we unfolded before, a first indication is surely the strengthening of the surplus population. However, we are not in a position to know the depth of the restructuring of the paternalistic type of class coalition between owners and workers, to the expense of the latter (finally the depth of the overthrow of class framework, which determined what we called particularities of the Russian private business economy and special historical field of expression of the crisis), which constitutes a precondition for the development of competition, except from the fact that the crisis itself weakens the negotiating power of the working class.

Second, in the state economic policy – as an expression of the class balance of power – which seems to be establishing an institutional environment of macroeconomic stability that permits the emergence of the main financial institutions (see O.E.C.D. 1997-8: 3, 149).

### 7. Conclusions

To sum up, the present paper used the *growth accounting* methodology to estimate technological change, in an attempt to formulate an explanation of Russia's economic decline and (signs of) recovery in the 1992-1999 time span, in relation to technological change. The results showed that, as a consequence of the economic collapse during the 1990s – rooted in the Soviet economy's very structure – the growth rate of the technological level remained practically unchanged, which, in turns limited the further deterioration of the Russian economy. Furthermore, the optimistic trends of the Russian economy towards recovery were discussed briefly.

<sup>&</sup>lt;sup>26</sup> Since real wages decrease and the family income also decreases, more family members enter the labour market looking for employment. Obviously, as long as the increase in labour supply exceeds the labour demand the rise in unemployment can go along with the rise of employment.

<sup>&</sup>lt;sup>27</sup> According to Kaitila (2003: 12), "the continuing existence of old state monopolies [...] is constraining growth". Enterprise structures change slowly, whereas market monopolisation has not been treated fully yet. An antimonopoly law in 1991 lacked effective enforcement, and the 1995 law on the regulation of natural monopolies despite being approved by the Duma, was vetoed by the President (O.E.C.D. 1995: 27 ff). Finally, the law "On competition and limitation of monopolistic activity on commodity markets" has been enforced only after some changes were introduced in 1998 (B.C.C.R. 2003). Most antimonopoly laws were adopted between 1991 and 1995 and are slowly being enforced.

The present paper also estimated potential output and output gap using the Cobb Douglas production function approach and the HP-filter technique. The Russian economy seems to be operating close to its potential. More precisely, potential GDP fell dramatically at 10% annually, while actual GDP fell at 8% rate annually. For the largest part of the 1992-1999 time span, actual output remained above the potential output, despite the decreasing pattern of both of them.

Furthermore, the paper examined the relationship between output gap and inflation. A positive gap for a time lag of three (3) years, where demand pressures on resources were present, tended to be associated strongly with increases in inflation. However, the short time series for the lag estimations reduced the capabilities of the statistical analysis. Also, the overall inflation rate is subject to external factors, such as depreciation rate of euro, high oil prices etc., and the output gap alone is not totally sufficient to explain inflation.

Our findings are, in general terms, consistent with the available theoretical and empirical literature. More precisely, the similar pattern that the natural and actual rates of unemployment follow is consistent with the findings in the seminal paper by Layard et al. (1991). Also, a rough comparison of the change in T.F.P. calculated and using the Growth Accounting methodology and the respective findings of the output gap methodological framework are absolutely consistent. Finally, the fact that output gap fuels inflation, or in other words the "domestically generated" character of the inflationary mechanism, which locates its origin inside the Russian social formation, and only partly in external factors, is another example of the general consistency of our results with the relevant literature (e.g. Milios 2001).

Closing, we would like to draw your attention to the fact that all estimates of output gap are subject to a margin error and the production function estimate is contingent on an estimate of the NAWRU to calculate potential employment. In the words of Arthur Okun: "The quantification of potential output – and the accompanying measure of the 'gap' between actual and potential – is at best an uncertain estimate and not a firm, precise measure". (Okun 1962, cited in Billmeier 2004, p. 3). The method we used is popular and appropriate, but it should be treated with caution in the Russian economy where a big share of shadow salary still exists. Obviously, future and more extended research on the subject would be of great interest.

As a conclusion, we may say that although a remarkable recovery has taken place in the Russian economy between 1998 and 1999, the dominance of monopolistic structures continues to exist, as the transition to more competitive structures is continuous but slow. Following Broadman (2001), it can be said that: "Despite privatization, robust competition is still lacking [...]. Many established firms enjoy protection from new [...] rivals. Reforming anticompetitive business structures and lowering barriers to entry are key to Russia's post-privatization reform"

In this framework, technical change and innovation are considered in Marx's perspective to emerge from the regularities determining the capitalist system as a whole, i.e. from the trends regulating the expanded reproduction of social capital, on the base of competition: Innovation and technical change are the main means of increasing labour productivity and "no less than other socio-economic activities, were best analysed as social processes" since "the focus of Marx's discussion of technology is [...] upon a collective, social process" (Rosenberg 1982: 35).<sup>28</sup>

<sup>&</sup>lt;sup>28</sup> Marx wrote: "A critical history of technology would show how little any of the inventions of eighteenth century are the work of a single individual" (Marx 1990: 493).

Consequently, production relations *per se* impose on *all individual capitals* the urge towards innovation and technical change. Capitalism cannot be technologically static, as much as it is identical with competition. Continuous innovation ensures on the one hand the increase in the rate of exploitation of labour by capital – and thus may raise the rate of profit – (what Marx describes, in Vol. 1 of *Capital* as "production of relative surplus-value"), while on the other it is the means *par excellence* for improving the individual enterprise's position vis-à-vis its competitors (Marx 1990: 959, 1037).

Conclusively, competitive market structures and enterprise restructuring are essential prerequisites for technological progress and economic growth. Following Milios *et al* (2002: 115): "The development of capitalism can be associated only with the evolution, not with the abolition, of free competition". Thus technological progress and sustainable economic growth in Russia still seem to "entail a long lasting process of [...] social rearrangements" (Milios 2001: 82-3). After all, crises are temporary destabilizations of the capitalist process of expanding reproduction and they also function as mechanisms that "re-establish the disturbed balance for the time being", as Marx (1991: 357) himself wrote.

#### References

Abramovitz, M., 1956, "Resource and Output Trends in the United States since 1870", *American Economic Review* 46, May, pp 5-23.

Analytis, M., 1999, "Structural Weaknesses and Prospects of the Oil Sector in Russia", in L. Maroudas and H. Tsardanidis (eds.), *Whither Russia? Economic and Political Dimensions of Transition*, Institute of International Economic Relations, Athens: Exantas, pp 293-312 (*in Greek*).

Andreff, W., 1978, "Structure de l'accumulation du capital et technologie en U.R.S.S.", *Revue d'études comparatives Est-ouest*, vol. IX, no1, Mars, pp 47-88.

Andrikopoulos, A., 2000, *Econometrics: Theory and Applications*, 2<sup>nd</sup> edition, Athens: Mpenos Publications (*in Greek*).

Artus, J.R. ,1997, Measures of Potential Output in Manufacturing for Eigh Indistrial countries, 1955-78, *Staff Papers*, International Monetary Fund, vol. 24, pp. 1-35.

Aukutsiovek, S., 1998, "Industrial barter in Russia", Communist Economies and Economic Transformation, Vol. 10, No 2, pp 179-188.

Baily, M. N. and Gordon, R. J., 1988, "The Productivity Slowdown, Measurement Issues, and the Explosion of Computer Power", *Brookings Papers on Economic Activity*, No 2, pp 347-420.

Billmeier, A. ,2004, "Ghostbusting: Which Output Gap Measure Really Matters?", *IMF Working Paper* WP/04/146.

Bolt, W. and P. J. A., van Els, Output Gap and Inflation in the E.U., *De Nederlandsche Bank Staff Report* No. 44.

Bolt, W. and van Els, P. J. A., 2000, "Output Gap and Inflation in the EU", *De Nederlandsche Bank Staff Report* No. 44.

Bowles, S. and Edwards, R., 1993, *Understanding Capitalism*, Harper Collins College Publishers.

Boycko, M., Shleifer, A. and Vishny, R. (1995), *Privitizing Russia*, Cambridge, Mass: M.I.T. Press.

Boycko, M., Shleifer, A. and Vishny, R., 1995, *Privitizing Russia*, Cambridge, Mass: M.I.T. Press.

Broadman, H., 2001, "Competition and Business Entry in Russia, Finance Development", *I.M.F. magazine* Vol. 38, No 2, June.

Business Cooperation Center Russia - A.P.E.C., 2003, *Russia: Competition Laws and Policies* and *The Ministry of the Russian Federation for Antimonopoly Policy and Support of Entrepreneurship* (http://apec.technet.ru/eng/).

Carchedi, G., 2001, For Another Europe, London, New York: Verso.

Chattopadhyay, P., 1990, "La dynamique de l'économie soviétique a la lumière de l'analyse marxienne de l'accumulation du capital", *Economie Appliquée*, tome XLIII, No 2, pp. 5-32.

Clarke, S., 1993, "The Contradictions of 'State Socialism". In: *What About the Workers? Workers and the Transition to Capitalism in Russia* (Clarke, S., Fairbrother P., Burawoy M., Krotov, P. eds.), London: Verso, pp 5-29.

Cogley, T. and Nason, J. (1995), Effects of the Hodrick – Prescott Filter on Trend and Difference Stationary Time Series: Implication for Business Cycle Research, *Journal of Economic Dynamics and Control*, vol. 19, pp. 253-78.

De Masi, P. (1997), I.M.F. Estimates of Potential Output: Theory and Practice, *I.M.F. Working Paper* 97/177.

Denison, E., 1967, Why Growth Rates Differ, Washington D.C.: The Brookings Institution.

Denison, E., 1985, *Trends in American Economic Growth*, 1929-1982, Washington, D.C.: The Brookings Institution.

Dimitz, M. A., 2001, "Output Gaps and Technological Progress in European Monetary Union", *Bank of Finland Discussion Paper*, No. 20, 2001

Dornbusch, R. and Fischer S., 1993, *Macroeconomics*, McGraw-Hill Publishing Company.

Doyle, P., Kuijs, L. and Jiang, G., 2001, "Real Convergence to EU Income Levels: Central Europe from 1990 to the Long Term", *I.M.F. Working Paper* WP/01/146.

Economakis, G., Markaki, M, Michaelides, P., Milios J. and Roboli, A. and Xenaki,

A. ,2004, "The ex-Soviet and the Russian Cargo and Tanker Fleet's Magnitude (1975-2001)", *East – West Journal of Economics and Business*, vol. VI, no 2, pp. 89-122.

Elmeskov, J. (1993), High and Persistent Unemployment: Assessment of the Problem and Its Causes, *O.E.C.D. Economics Department Working Paper*, No. 132.

European Bank for Reconstruction and Development, 1999, *Transition Report*. Express, 1996, *Russia*, Internet Edition (*Greek daily newspaper*).

Fagan, G., Henry, J. and Mestre, R. (2001), An Area – Wide Model (AWM) for the Euro Area: Estimates and assessment, *ECB Working Paper* No. 17.

Fakiolas, A. and Fakiolas, R., 1999, *The Economic Evolution in the Russian Federation: Difficulties in the Establishment of the Market* in L. Maroudas and H. Tsardanidis

(eds.), Whither Russia? Economic and Political Dimensions of Transition, Institute of International Economic Relations, Athens: Exantas, pp.179-200 (in Greek).

Filtzer, D., 1994, Soviet Workers and the Collapse of Perestroika: The Soviet labour process and Gorbachev's reforms, 1985-1991, New York: Cambridge University Press.

Gavrilenkov, E. and Koen, V., 1995, "How Large was the Output Collapse in Russia? Alternative Estimates and Welfare Implications", *Staff Studies for the World Economic Outlook*, Washington DC: Fonds Monetaire International, pp. 59-67.

Gimpelson, V. and Lippoldt, D. (1997), Labour Turnover in the Russian Economy, in: *Labour Market Dynamics in the Russian Federation*, O.C.D.E.

Giorno, C., Richardson, P., Roseveare, D. and van der Noord, P. (1995), Estimating Potential Output, Output Gaps and Structural Budget Balances, *O.E.C.D. Working Papers*, No 152.

Gorbachev, M., 1987, *Persestroika, A New Thought for Our Country and the World*, Athens: Nea Synora Publications (*in Greek*).

Goskomstat, 1997, Russia in Figures, Moscow.

Griliches, Z., 1988, "Productivity Puzzles and R&D: Another Nonexplanation", *Journal of Economic Perspectives* 2, Fall, pp. 9-21.

Haaparanta, P. and Leena, K., 2000, *Transition and Barter in Russia*, ETSG Papers, August (http://www.etsg.org/ETSG2000/Papers/Kerkela.pdf.)

Harvey, A.C., and Jaeger, A. (1993), Detrending, Stylized Facts and the Business Cycle, *Journal of Applied Econometrics*, Vol. 8, pp. 231-47.

International Monetary Fund, 1994, World Economic Outlook, Washington DC.

International Monetary Fund, 2001, 2002a, 2003, 2004, *International Financial Statistics*, Washington DC.

International Monetary Fund, 1994, World Economic Outlook, Washington DC.

International Monetary Fund, 2001, 2002a, 2003, 2004, *International Financial Statistics*, Washington DC.

International Monetary Fund, 2002b, Russian Federation: Selected Issues and Statistical Appendix, Country Report No 02/75, April, Washington DC.

Jorgenson, D., 1988, "Productivity and Postwar U.S. Economic Growth", *Journal of Economic Perspectives* 2 (Fall), pp 23-41.

Kagarlitsky, B., 1995, Restoration in Russia. Why Capitalism Failed, London-New York: Verso.

Kaitila, V., 2003, "An Assessment of Russia's Growth Prospects in 2003-2010", The Research Institute of the Finnish Economy, *Discussion Paper* No 869. September.

Kapeliouchnikov, N., 1997, "Job and Labor Turnover in Russian Industry", Russian Economic Barometer, No 1.

Kapelyushnikov, R.I., 2001, Rossiiskii rynok truda: adaptatsiya bez restrukturizatsii, GU VShE, Moscow.

Kenny, G. (1996), Capacity Utilization in Irish Manufacturing, Central Bank of Ireland Technical Paper Series, No. 2/RT/96.

King, R.G. and Rebelo, S. (1993), Low Frequency Filtering and Real Business Cycles, *Journal of Economic Dynamics and Control*, Vol. 17, pp. 361-8.

Kuboniwa, M. and Gavrilenkov, E., 1997, *Development of Capitalism in Russia: The Second Challenge*, Tokyo: Maruzen Co.

Kuboniwa, M., 1996, *Economic Growth in Postwar Russia: Estimating GDP*, Institute of Economic Research, Hitotsubashi University, Document de discussion No D96-10.

Kuttner, K. (1994), Estimating Potential Output as a Latent Variable, Journal of Economics and Business Statistics 12, pp. 361-368.

Layard, R., Nickell, S., Jackman, R. (1991), Unemployment, Macroeconomic Performance and the Labour Market, Oxford University Press.

Liodakis, G., 2001, "The Political Economy of the Transition and the Transformation Crisis in Eastern Europe", *East – West Journal of Economics and Business* Vol. IV, No 2, pp 45-64.

Loasby, B. J., 2002, "Evolution and Institutions: A Cognitive Perspective", *East – West Journal of Economics and Business* Vol. V, No1, pp 27-45.

Mankiw, N. G., Romer, D. and Weil, D., 1992, "A Contribution to the Empirics of Economic Growth", *The Quarterly Journal of Economics*, May, pp 407-437.

Marangos, J., 2003a, "A Political Economy Methodology in Modelling the Transition Process", *Global Business & Economics Review*, Vol.5, No. 2, pp. 197-225.

Marangos, J, 2003b, "Global Transition Strategies in Eastern Europe: Moving to Market Relations", *Development*, Vol. 46, No. 1, pp. 112-7.

Marangos, J., 2004, "Modelling the Privatization Process in Transition Economies", *Oxford Development Studies*, Vol. 32, No. 4, pp. 585-604.

Maroudas, L. and Rizopoulos, Y., 2002, "The Russian Firm's Organisational Links and Behaviour", *East – West Journal of Economics and Business* Vol. V, No 2, pp 123-137.

Maroudas, L. E., 2001, "Organizational Changes and Human Resource Management in Post-Soviet Industrial Enterprises (1990-1995)", *East – West Journal of Economics and Business* Vol. IV, No 1, pp 51-70.

Marx, K., 1990, Capital, Vol. 1, London: Penguin.

Marx, K., 1991, Capital, Vol. 3, London: Penguin.

Michaelides, P, Economakis, G. Milios, J., Maroudas, L. and Aggelis, V. (2005), Growth and Technological Change in the Russian Economy: A Contribution to the Investigation of Russia's Economic Crisis, *East – West Journal of Economics and Business*, October (in print).

Milios J., 1997, Modes of Production and Marxist Analysis, Athens: Kritiki (in Greek).

Milios, J., 2001, "Investigating The Roots Of Russia's Economic Decline Since 1990", *East-West Journal of Economics and Business*, Vol. IV, No 2, pp 65-84.

Milios, J., Dimoulis, D. and Economakis, G., 2002, Karl Marx and the Classics, Aldershot: Ashgate.

Mokyr, J., 1990, *The Lever of Riches, Technological Creativity and Economic Progress*, New York: Oxford University Press.

Nove, A., 1977, The Soviet Economic System, London: George Allen & Unwin Ltd.

O.C.D.E., 1997, "Fédération de Russie". In : *Etudes Economiques de l' O.C.D.E.* 1997, Paris.

O.C.D.E., 1997, "Fédération de Russie". In : *Etudes Economiques de l' O.C.D.E.* 1997, Paris

O.E.C.D., 1991, The Transition to a Market Economy, Paris.

O.E.C.D., 1995, Economic Surveys, The Russian Federation, Paris.

O.E.C.D., 1996, Research and Development Expenditures in Industry, 1973-1993, Paris.

O.E.C.D., 1997-8, Economic Surveys, 1997-1998, Paris.

O.E.C.D., 2003, Main Economic Indicators, Paris (various years)

Okun, A. (1962), Potential G.N.P.: Its Measurement and Significance, *Proceedings of the Business and Economics Section*, American Statistical Association, pp. 98-104.

Page, J., 1994, "The East Asian Miracle: Four Lessons for Development Policy", *NBER Macroeconomics Annual* 9, pp 219-269.

Pavlovich, A. A., 1996, *Long Term Economic Growth Strategy*, Working Paper No 47, Economic Research Institute, Tokyo: Agence de Planification Economique.

Radyguine, A., 1997, Novaia faza raspredeleniia sobstvennosti, *Rossiiskaia ekonomika: tendentsii I perspektivy*, Institut pour l' etude des problemes de l' economie en trasition (koi.www.online.ru/sp/trends/mai97).

Romer, D., 1996, Advanced Macroeconomics, The McGraw Hill Companies.

Rosenberg, N., 1982, *Inside the Black Box, Technology and Economics*, Cambridge University Press.

Rutland, P., 1985, The Myth of the Plan, London: Hutchinson and Co.

Samouel, P., Kalafatis, S., Browne, D., Skok, W., Sims, T. and Rowe, D. (1996), *Information Management*, Master of Business Administration, ICBS, Kingston University Press.

Senhadji, A. (2000), Sources of Economic Growth: An Extensive Growth Accounting Exercise, *I.M.F. Staff Paper* Vol. 47, No. 1.

Siriopoulos, C. and Asteriou, D., 2001, "Emerging Financial Systems in Central And Eastern European Countries", *East – West Journal of Economics and Business* Vol. IV, No 2 pp 85-97.

Slevin, G., 2001, "Potential Output and The Output Gap in Ireland", Central Bank of Ireland Technical Paper, 5/RT/01, September 2001.

Solow, R., 1957, "Technical Change and the Aggregate Production Function", *Review of Economics and Statistics*, 39, 3, pp 312-320.

Staiger, D.; Stock J. H.; Watson M. W. (1997), The NAIRU, Unemployment and Monetary Policy, *Journal of Economic Perspectives*, Vol. 11, No. 1, 1997, pp. 3-10.

Stikuts, D., 2003, "Measuring output gap in Latvia", *Latvijas Banka*, Riga, Working Paper 2/2003 - 22 S.

Thirlwall, A., 2001, Growth and Development, Athens: Papazissis Publications (in Greek).

Torres, R., Martin, J., P. (1990), Measuring Potential Output in the Seven Major O.E.C.D. Countries, O.E.C.D. Economic Studies No. 14, Spring.

United Nations, 1996, The Economic Survey for Europe 1995-96, Internet Edition.

Young, A., 1994, "The Tyranny of Numbers: Confronting the Statistical Reality of the East Asian Growth Experience", *NBER Working Paper* No 4680, March.

Tsolas, J., 1995, "A Study of Productivity in the coal mining sub-sector", *Thesseis*, 53 (*in Greek*).

## Appendix

The Fixed Capital Formation in the Russian Economy was estimated using the *Perpetual Inventory method* and the Production Prices Index (I.M.F. 2002a, 2004). The equations used are the following:

K(i) = [S(i) + S(i-1)]/2  $S(i) = \sum_{t=1}^{i} I(t)$  I(t) = E(t) - d(t) E(t) = E'(t)/p(t) $d(t) = E(t)[\gamma \cdot (\gamma + 1)^{t-1}]/[(1+\gamma)^{1+T} - 1]$ 

where:

K(i): fixed capital in the middle of year i (constant prices)

S(i): fixed capital at the end of year i (constant prices)

I(t): net fixed capital investments (constant prices)

E(t): gross fixed capital investments (constant prices)

E'(t): gross fixed capital investments (current prices)

p(t): production prices (index)

d(t): consumption of fixed capital depending on the depreciation policy

 $\gamma = 0.20$  parameter depending on the time-structure of investments (Tsolas 1995)

T=15 the average lifetime of machinery and equipment (Goskomstat 1997).